Author's response to reviews

Title: What is the global burden of visual impairment?

Authors:

Lalit Dandona (dandona@asci.org.in)
Rakhi Dandona (rakhi@asci.org.in)

Version: 2 Date: 2 December 2005

Author's response to reviews:

This paper was submitted yesterday. Some errors were detected, which are corrected in today's version of this paper.